Author’s response to reviews

Title: Impact of NPS MedicineWise general practitioner education programs and Choosing Wisely Australia recommendations on prescribing of proton pump inhibitors in Australia

Authors:

Jianyun Wu (jianyun.fred.wu@gmail.com)
Scott Dickinson (scott.dickinson@saxinstitute.org.au)
Zain Elgebaly (ZElgebaly@nps.org.au)
Suzanne Gaye Blogg (sblogg@nps.org.au)
Aine Heaney (AHeaney@nps.org.au)
Yien Soo (yien.soo@gmail.com)
Benjamin Daniels (b.daniels@unsw.edu.au)
Lynn Weekes (lweekes2070@gmail.com)

Version: 4 Date: 30 Apr 2020

Author’s response to reviews:

--Declaration sub-sections must be structured into the following sections:

- Ethics approval and consent to participate
- Consent for publication
- Availability of data and material
- Competing interests
- Funding
- Authors' contributions
- Acknowledgements

Please revise your Declaration structure to conform to the journal's submission guidelines.
Completed
-- Please remove the funding information from the Acknowledgements and include it in the Funding section instead. If you have no further acknowledgements please put “Not Applicable” in the Acknowledgements section.

Acknowledged supply of data and staff who worked on the project.